Author's response to reviews

Title: Evaluation of effectiveness of instruction and study habits in two consecutive clinical semesters of the Medical Curriculum Munich (MeCuM) reveals the need for more time for self study and higher frequency of assessment

Authors:

Sophia Mueller (sophia.mueller@med.uni-muenchen.de)
Nina Weichert (nina.weichert@gmx.de)
Veit Stoecklein (veit.stoecklein@med.uni-muenchen.de)
Ariane Hammitzsch (ariane.hammitzsch@med.uni-muenchen.de)
Giulia Pasciuto (giulia.mosi@inwind.it)
Christian Krug (christian.krug@med.uni-muenchen.de)
Matthias Holzer (matthias.holzer@med.uni-muenchen.de)
Mona Pfeiffer (mona.pfeiffer@med.uni-muenchen.de)
Matthias Siebeck (matthias.siebeck@med.uni-muenchen.de)
Ralf Schmidmaier (ralf.schmidmaier@med.uni-muenchen.de)

Version: 4 Date: 24 August 2011

Author's response to reviews:

To the Editorial Production Team:

according to your advice we made the following changes to our manuscript:

1. Affiliations are now written in full
2. Abstract should now be the same in manuscript and submission system
3. figure 1A and 1B are now figure 1 and 2
4. tracking is erased
5. authors' contributions are formatted as paragraph
6. figures are cropped as closely as possible

Please contact us for any further advice or suggestions,

thank you very much for your assistance,

Sophia Mueller